Dear reviewers and Members of the Editorial Board,

We would like to submit a revised original research manuscript by Goran Livazović and Karla Kuzmanović entitled “Predicting adolescent perfectionism: the role of socio-demographic traits, personal relationships and media” to World Journal of Clinical Cases for possible evaluation and publication. It is a theoretical and empirical research paper on the relation between perfectionism and the quality of family, peer and college relationships and media usage and content interests, which we find may be of interest to the readers of Your journal. Our paper was previously evaluated by the World Journal of Psychiatry, and recommended for publishing in the World Journal of Clinical Cases.

We believe the research topic is in accordance with the aim and scope of the World Journal of Clinical Cases. After a thorough analysis of the Guidelines for Authors and the reviewers’ recommendations, we have adjusted the technical, theoretical and empirical elements of the paper accordingly.

The References section of the paper was adjusted according to the reviewers’ suggestions, and the text was revised language-wise.

Correspondence and phone calls about the paper please be directed to Goran Livazović, at the following address, telephone and fax number, and e-mail address:

Goran Livazović, PhD, Associate Professor

Department for Pedagogy

Faculty of Humanities and Social Sciences

University of J. J. Strossmayer in Osijek
We affirm that this manuscript is original, has not been published before and is not currently being considered for publication elsewhere.

Thank You very much for Your attention to our paper.

Goran Livazović          Karla Kuzmanović